Author's response to reviews

Title: Ardisia crispa roots inhibit cyclooxygenase and suppress angiogenesis

Authors:

Dayang Erna Zulaikha Awang Hamsin (dayangerna@hotmail.com)
Roslida Abdul Hamid (roslida@medic.upm.edu.my)
Latifah Saiful Yazan (latifah@medic.upm.edu.my)
Che Norma Mat Taib (norma@medic.upm.edu.my)
Yeong Looi Ting (joyce_ylt86@hotmail.com)

Version: 3
Date: 17 January 2014

Author's response to reviews: see over
Dear Editor

I am attaching our corrected manuscript herewith for your kind consideration. We would like to thank reviewers for their constructive advices and comments. The followings are the corrections that we have made as per reviewer’s suggestions;

Reviewer: Dr Nafiseh Khosravi

1. In plant material part, correction has been made according to suggestion given. “…n-hexane (2000 ml x 3, 48 hr each).
2. Result and discussion part have been further summarized, as per suggestion.

For the suggestion to replace old reference with the latest one, we believe that change might not be applicable to some the references as some of the reference is still important to be cited in this manuscript. For instance, Jansakul et al (1987) first discovered Ardisiacrispin A and B (Reference no 15). We believe it is important to cite these researchers even though the reference might be outdated.

We hope that these corrections are sufficient for our manuscript to be published. We look forward to hearing from you soon. Last but not least, we thank you for your thoughtful consideration and attention.

Best regards,
Dayang Erna Zulaikha
Pharmacology Unit
Faculty of Medicine and Health Sciences
Universiti Malaysia Sarawak
Malaysia.